Author's response to reviews

Title: Assessment of stereoscopic optic disc images using an autostereoscopic screen. experimental study.

Authors:

Maged S Habib (mhabib jc@yahoo.co.uk)
James A Lowell (james.a.lowell@gmail.com)
Nick S Holliman (n.s.holliman@durham.ac.uk)
Andrew Hunter (ahunter@lincoln.ac.uk)
Daniella Vaideanu (daniela.vaideanu@newcastle.ac.uk)
Anthony Hildreth (tony.hildreth@chs.northy.nhs.uk)
David H Steel (david.steel@chs.northy.nhs.uk)

Version: 4 Date: 22 July 2008

Author's response to reviews:

Dear Sir
The article has been checked and formatted for publication
1. Title was sentenced (capitals removed)
2. Authors list had been checked, emails written in title page in the proper format.
3. References been formated as sequential numbers in square brackets.
4. Authors contribution section was completed
5. the article was checked for any spelling mistakes or typographical errors.
Hope these changes and adjustments are satisfactory for publication

Many thanks
Kind regards
Maged Habib